

Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2023GL107613

Key Points:

- Considering a sufficient land depth in current-generation Earth system models facilitates realistic land heat uptake estimates under warming
- Increased land heat uptake leads to an adjusted global heat distribution and a reduced overestimation of simulated ocean heat uptake
- The land heat sink is relevant for the projection of the global energy inventory and its partitioning under global warming

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

N. J. Steinert, normanst@ucm.es

Citation:

Steinert, N. J., Cuesta-Valero, F. J., García-Pereira, F., de Vrese, P., Melo Aguilar, C. A., García-Bustamante, E., et al. (2024). Underestimated land heat uptake alters the global energy distribution in CMIP6 climate models. *Geophysical Research Letters*, *51*, e2023GL107613. https://doi. org/10.1029/2023GL107613

Received 30 NOV 2023 Accepted 12 APR 2024

Author Contributions:

Conceptualization: Norman Julius Steinert, Francisco José Cuesta-Valero, Félix García-Pereira, Philipp de Vrese, Camilo Andrés Melo Aguilar, Elena García-Bustamante, Johann Jungclaus, Jesús Fidel González-Rouco

Formal analysis: Norman Julius Steinert Investigation: Norman Julius Steinert, Francisco José Cuesta-Valero, Félix García-Pereira Methodology: Norman Julius Steinert, Francisco José Cuesta-Valero, Félix García-Pereira Supervision: Philipp de Vrese, Elena García-Bustamante.

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Underestimated Land Heat Uptake Alters the Global Energy Distribution in CMIP6 Climate Models

Norman Julius Steinert^{1,2}, Francisco José Cuesta-Valero^{3,4}, Félix García-Pereira¹, Philipp de Vrese⁵, Camilo Andrés Melo Aguilar^{1,6}, Elena García-Bustamante⁷, Johann Jungclaus⁵, and Jesús Fidel González-Rouco¹

¹Geosciences Institute IGEO (UCM-CSIC), Complutense University of Madrid, Madrid, Spain, ²NORCE Norwegian Research Centre, Bjerknes Centre for Climate Research, Bergen, Norway, ³Department of Remote Sensing, Helmholtz Centre for Environmental Research – UFZ, Leipzig, Germany, ⁴Remote Sensing Centre for Earth System Research, Leipzig University, Leipzig, Germany, ⁵Max Planck Institute for Meteorology, Hamburg, Germany, ⁶Balearic Islands Oceanographic Centre, Palma, Spain, ⁷Research Center for Energy, Environment and Technology (CIEMAT), Madrid, Spain

Abstract Current global warming results in an uptake of heat by the Earth system, which is distributed among the different components of the climate system. However, current-generation climate models deliver heat inventory and partitioning estimates of Earth system components that differ from recent observations. Here we investigate the global heat distribution under warming by using fully-coupled CMIP6 Earth system model experiments, including a version of the MPI-ESM with a deep land model component, accommodating the required space for more realistic terrestrial heat storage. The results show that sufficiently deep land models exert increased subsurface land heat uptake, leading to a heat uptake partitioning among the Earth system components that is closer to observational estimates. The results are relevant for the understanding of Earth's heat partitioning and highlight the importance of the land heat sink in the Earth heat inventory.

Plain Language Summary Global warming is associated with heat accumulation in the Earth system due to the intensification of the greenhouse effect. The available heat is distributed unevenly throughout the climate subsystems: the ocean, land, atmosphere, and cryosphere. Overall, the current generation of climate models captures this partitioning well but, on average, shows an overestimation of the ocean heat uptake and an underestimation of the land heat uptake. Previous studies have shown that the lack of heat input into the land comes from structural limitations in the land model components used. In this study, we account for these shortcomings, which greatly improve the land heat uptake in simulations of future climate scenarios. We find that, as a result, the fraction of simulated heat taken up by the ocean is reduced. This leads to a heat distribution among the climate subsystems that is closer to observational estimates. Our results highlight that land heat uptake is relevant for the Earth system heat distribution and that future research should consider modeling approaches including a more realistic land heat sink.

1. Introduction

"Where does the energy go?" is a long-standing question in climate research that addresses the amount of heat uptake by the different components of the Earth system as a response to a positive global energy imbalance (von Schuckmann et al., 2020, 2023). Currently, the Earth system is in a positive radiative imbalance due to anthropogenic emissions of greenhouse gases (Hansen et al., 2005, 2011), which gradually increases its heat storage. For the period of 1971-2020, 381 ± 61 ZJ were observed to be stored in the Earth system, accounting for ocean (89%), land (6%) and atmosphere (1%) warming, and cryosphere melting (4%) (Cuesta-Valero et al., 2016, 2021, 2023; von Schuckmann et al., 2020, 2023).

Determining the distribution of this heat among the climate system components is important because the heat storage of different Earth system components and the transfer between them largely dictates the state and variations of the global climate (Hansen et al., 2011). For example, heat uptake within these subsystems drives variations in thermodynamic and hydrological cycles (Rhein et al., 2013) and has an impact on the global carbon cycle, ecosystem functioning, agricultural production, environment, and society (Gornall et al., 2010).



Johann Jungclaus, Jesús Fidel González-Rouco

Visualization: Norman Julius Steinert Writing – original draft: Norman Julius Steinert, Francisco José Cuesta-Valero, Félix García-Pereira Writing – review & editing: Francisco José Cuesta-Valero, Félix García-Pereira, Philipp de Vrese, Camilo Andrés Melo Aguilar, Elena García-Bustamante, Johann Jungclaus, Jesús Fidel González-Rouco Earth system models (ESMs) are employed to estimate the evolution of Earth's heat uptake and its distribution in the future. Cuesta-Valero et al. (2021) provide the first complete assessment of the heat distribution in ESMs participating in the Climate Model Intercomparison Project (CMIP) Phase 5. They find a fractional ocean heat uptake of 96% \pm 4%, land warming of 2% \pm 3%, and accumulation of heat in the atmosphere and cryosphere at $1\% \pm 1\%$, respectively, of the total system heat uptake. These results represent an overestimation of the amount of ocean heat storage and an underestimation of heat uptake by the cryosphere and land components in comparison to the observational estimates (Cuesta-Valero et al., 2021). Previous literature has suggested that land heat uptake under climate change conditions in ESMs is compromised by too shallow land surface model components (LSMs) because a subsurface bottom boundary condition too close to the surface biases the representation of subsurface heat propagation and heat distribution (Alexeev et al., 2007; Cuesta-Valero et al., 2016, 2021; González-Rouco et al., 2009, 2021; MacDougall et al., 2008; Smerdon & Stieglitz, 2006; Steinert et al., 2024; Steinert, González-Rouco, de Vrese, et al., 2021; Steinert, González-Rouco, Melo-Aguilar, et al., 2021; Stevens et al., 2007). Despite recent improvements in modeling land processes in ESMs (Fisher & Koven, 2020), only limited attention has been directed toward the effect of land model depth and its impact on the representation of terrestrial thermodynamics. Therefore, the bulk of the current generation of LSMs within ESMs still has depths that range between 3 and 14 m (Burke et al., 2020; Cuesta-Valero et al., 2016), allocating limited space for subsurface heat propagation and hence, land-climate feedbacks (González-Rouco et al., 2009). The exceptions are ESMs using the Community Land Model (Lawrence et al., 2019; Oleson et al., 2013) and the ORCHIDEE model (Cheruy et al., 2020), extending their bottom boundaries down to around 45 and 90 m, respectively, which leads to a higher and more realistic land heat storage (Cuesta-Valero et al., 2021).

While the impacts of realistically deep land components have been investigated with standalone LSMs and analytical models, they have not yet been considered in the frame of ESM coupled simulations (Phillips, 2020). Here, we hypothesize that ESMs with deeper land model components not only show larger land heat storage but are consequently adjusting the distribution of excess heat among the climate system components toward fractions closer to observational estimates. We therefore explore the influence of a deepened land model component considering an ensemble of existing CMIP6 climate model simulations and support our analysis by a current-generation ESM that we modified to encompasses both a deep and a shallow land model setup. We consider the land model depth of CMIP6 ESMs in the assessment of the Earth system heat inventory and investigate the effects of a more realistic land heat sink on the global heat distribution.

2. Model Data and Methodology

2.1. CMIP6 Ensemble

Simulations of 33 CMIP6 ESMs are considered for our analysis of the Earth energy inventory (Table S1 in Supporting Information S1). Their land model depth varies between 3 and 90 m, with 61% lying between 3 and 14 m. Some ESMs use an identical land model coupled with different ocean or atmosphere components. Only one ensemble member per model is considered. For our analysis, we use the historical (1850–2014) and the future (2015–2100) SSP5-85 forcing scenario periods (Gidden et al., 2019; Meinshausen et al., 2017). Additionally, the climate trends of all available years of the piControl simulations were used to detrend the transient CMIP6 climate projections. This is a necessary step to remove spurious drifts from incomplete spin-up processes in global climate models because ESMs require millennia under stable conditions to reach an equilibrium state, mostly due to the long timescales of the global ocean circulation (Gupta et al., 2013; Peacock & Maltrud, 2006; Séférian et al., 2016). Previous analyses have shown that this dedrifting technique mitigates the non-conservation terms in the simulated Earth heat inventory within transient climate simulations (Hobbs et al., 2016; Irving et al., 2020), which is crucial for our analysis. For the calculation of the Earth system heat uptake and distribution (see Section 2.4), we use a total of 20 output variables provided by the CMIP6 protocol that are summarized in Table S2 of Supporting Information S1.

2.2. MPI-ESM Ensemble

CMIP6 models provide outputs for the LSM component with a single depth configuration. The large inter-model variability in global heat uptake and non-conservation of energy within the models (Irving et al., 2020; Wilde & Mulholland, 2020) is thus challenging to attribute to differences in land model depth. Therefore, we support our multi-model ensemble with coupled simulations of the Max Planck Institute for Meteorology Earth System Model

(MPI-ESM; Mauritsen et al., 2019) with two different LSM setups. The first setup equals the standard CMIP6 version of MPI-ESM, which uses the JSBACH LSM (Reick et al., 2021) with five layers (9.83 m; SHALLOW hereafter). For the second setup, we introduce modifications to the vertical structure of JSBACH, which uses an extended 18-layer discretization with a larger model depth (1,417 m; DEEP hereafter) that is well suited for the simulation of centennial to millennial timescales (González-Rouco et al., 2021; Steinert et al., 2024; Steinert, González-Rouco, de Vrese, et al., 2021; Steinert, González-Rouco, Melo-Aguilar, et al., 2021). The two simulations were prepared by restarting the ocean component MPIOM from an existing CMIP6 control simulation. Prior to the latter, a spin-up run was conducted over several centuries starting from the Polar Science Center ocean hydrographic climatology (Steele et al., 2001). Using an identical fully coupled model with two different depth configurations reduces the uncertainties associated with model differences in simulating the global energy budget.

2.3. Heat Conduction Forward Model

The dependence of the land heat uptake on the depth of the LSMs is evaluated using an offline one-dimensional numerical heat conduction forward model (HFMod; García-Pereira et al., 2024). This model numerically discretizes the vertical heat conduction equation (Carslaw & Jaeger, 1959):

$$\frac{\partial T}{\partial t} = \kappa \frac{\partial^2 T}{\partial z^2},\tag{1}$$

where *T* represents the temperature at a specific depth *z* and time step *t*, and κ denotes the thermal diffusivity. The numerical solution is attained using a Finite Time Centered Space finite difference scheme, consisting of an explicit Euler scheme for time resolution and a second-order central difference for the spatial dimension (Roache, 1998). The resulting numerical solution of the heat conduction equation is expressed as:

$$T_i^{n+1} = T_i^n + \kappa \Delta t \, \frac{T_{i-1}^n - 2T_i^n + T_{i+1}^n}{\Delta z^2},\tag{2}$$

with T_i^n representing the temperature value at soil level *i* during time step *n*. The model produces the subsurface vertical temperature structure at each time step, employing a thermal equilibrium boundary condition at the ground surface, that is, $T[z = 0] = T_0$, and a zero-flux boundary condition at the bottom, that is, $T[z = z_{max}] = 0$. To ensure the stability of the numerical scheme, governed by the Courant-Friedrichs-Lewy condition $\Delta t \le \Delta x^2/2\kappa$ (Courant et al., 1928), a constant time step of 1 day and a soil layer thickness of 1 m was employed. The zero-flux bottom boundary condition was imposed at an infinite depth (HFMod infinite solution) and specific depths (HFMod limited solution).

2.4. Earth Energy Calculations

Earth system heat content estimates were obtained from CMIP6 outputs following the methods described in Cuesta-Valero et al. (2016, 2021). The data used for these calculations is summarized in Table S2 of Supporting Information S1. Ocean heat content (OHC) integrates the simulated temperature and salinity profiles using the algorithm for potential enthalpy (Griffies, 2004; McDougall, 2003). To convert the retrieved potential enthalpy into heat content, seawater density and pressure are estimated for each ocean layer as functions of depth, temperature, and salinity (McDougall et al., 2003; Smith et al., 2010).

Land heat content (LHC) is estimated by integrating the simulated subsurface temperature and water profiles for each model. Volumetric heat capacity is derived from sand, clay, water content, and ice content at each model layer (Oleson et al., 2010; Van Wijk et al., 1963), with sand and clay data retrieved from the ECOCLIMAP project (Champeaux et al., 2005). Simulated soil water and ice content at each layer are classified as water if the temperature of the layer is 0°C or above and ice otherwise. For simulations providing only column-integrated soil data, the water and ice contents are distributed proportionally within the soil layers above the bedrock and then classified as water or ice as in the layer-resolved cases (Cuesta-Valero et al., 2016, 2021). The thermal properties of bedrock are considered to start at 3.8 m and expand until the bottom layer of the model for all models to have a common depth reference, even for models that do not define bedrock (including the MPI-ESM simulations used herein). The volumetric heat capacity for bedrock is assumed to be $\rho_C = 3.0 \cdot 10^6$ Jm⁻³ K⁻¹ (Cuesta-Valero et al., 2016, 2021; MacDougall et al., 2008, 2010).



Geophysical Research Letters



Figure 1. (a) Global average land surface temperature (excluding glaciers) and (b) land heat uptake in 33 ESMs participating in CMIP6 (Table S1 in Supporting Information S1) for the period 1850–2100 relative to pre-industrial conditions. The blue and gray scales stand for model ECS in (a) and model depth in (b), respectively. (c) Global average cumulative land heat uptake (land heat uptake sensitivity; excluding glaciers) in relation to equilibrium climate sensitivity (bubble size; Table S1 in Supporting Information S1). Bubble colors in panels (c) and (d) refer to the respective model depths as indicated in the legend in panel (b). (d) Global average cumulative land (excluding glaciers) heat uptake per degree of global land surface temperature change (land uptake efficacy) with land model depth (Table S1 in Supporting Information S1) for the period of 1850–2100 relative to pre-industrial conditions. Shading indicate the range of results from the heat conduction forward model with the soil lines representing values of $1.0 \cdot 10^{-6} \text{ m}^{-2} \text{ s}^{-1}$ for the thermal diffusivity, and $3.0 \cdot 10^{6} \text{ Jm}^{-3} \text{ K}^{-1}$ for the volumetric heat capacity (Figure S2 in Supporting Information S1).

Several variables are used to derive atmospheric heat content (AHC). The simulated atmospheric profiles of temperature, wind, and specific humidity are integrated following the theory in Trenberth (1997) and the practical method described in Previdi et al. (2015), in combination with simulated values of surface pressure and geopotential (Dutton, 2002). The specific heat capacity and latent heat of vapourization are considered constant for the entire atmospheric profile, with values of $c_p = 1,000 \text{ Jkg}^{-1} \text{ K}^{-1}$ and $L_v = 2,260 \text{ Jkg}^{-1}$, respectively.

Regarding the cryosphere heat content (CHC), simulated changes in ice masses are considered, with model outputs allowing us to estimate ice melt in different components. All but three models provide enough variables to estimate changes in sea ice mass, with a varying number of models presenting enough variables to estimate changes in subsurface ice and snow mass (Table S2 in Supporting Information S1). None of the models explicitly simulate changes in glacier extent or the ice caps in Greenland and Antarctica, a known limitation of CMIP models (Cuesta-Valero et al., 2021; Flato et al., 2013). The heat content in the different simulated components of the cryosphere is estimated as the change of mass multiplied by the latent heat of fusion $L_f = 3.34 \cdot 10^5$ Jkg⁻¹ (Rhein et al., 2013).

3. Land Heat Uptake Across CMIP6 Models

The land surface temperature (here defined as Ground Surface Temperature—the first model layer) in the CMIP6 ensemble shows a pronounced increase over the 20th and 21st centuries, following the imposed socio-economic pathway SSP5-85 (Figure 1a). The spread among the 33 participants ranges between about 4 and 10 K at the end

of the 21st century. With the land heat uptake directly related to the increase in surface temperature, the former is expected to increase accordingly. However, the land energy uptake among the CMIP6 models show a surprisingly large spread of 6–143 ZJ by 2100 (Figure 1b). Higher land surface temperatures do not lead to a proportional increase in land heat uptake in many CMIP6 models (Figure 1c). This is caused by differences in land model depth (Figure 1b). Equilibrium climate sensitivity (ECS; Meehl et al., 2020; Schlund et al., 2020; Zelinka et al., 2020, Table S1 in Supporting Information S1) can be excluded to cause the range of land heat uptake for the different CMIP6 models because the spread persists when accounting for the ECS (Figure 1c; bubble size). The modeldepth dependency of the land heat uptake leads to a notable separation into two groups in the CMIP6 land heat content (Figures 1b and 1c). This separation motivates a split of the CMIP6 ensemble into shallow and deep model groups according to their respective land model depth (Table S1 in Supporting Information S1). Shallower models between 3 and 14 m depth (Figure 1c, orange) show no significant relationship between land heat uptake and surface warming ($R^2 = 0.00$). These models are limited to 31 ZJ (6–93 ZJ; indicating minimum and maximum values) heat uptake between 1850 and 2100, despite some of them simulating up to 9.2 K of global mean temperature increase. Contrarily, the group of deeper models between 42 and 90 m (Figure 1c, turquoise) correlate linearly with the amount of surface warming ($R^2 = 0.86$), with heat uptake of 101 ZJ (56–143 ZJ). The two extra MPI-ESM setups fall well into these two depth categories (Figure 1c). As MPI-ESM has a small ECS (Meehl et al., 2020; Schlund et al., 2020; Zelinka et al., 2020), the amount of warming and respective energy uptake is on the lower end of the CMIP6 ensemble.

To remove the influence of differences in land surface warming across ESMs, we define land heat uptake efficacy (LHUe) to be the land heat uptake divided by the surface warming (Figure 1d). The LHUe shows large differences between the shallow and the deep CMIP6 model groups. Generally, deeper land models equal larger land energy uptake. However, even the deepest CMIP6 models are shallower than 170 m, which was identified as the recommended depth to fully resolve the propagation of the surface temperature signal in historical and climate change scenario experiments (Steinert, González-Rouco, Melo-Aguilar, et al., 2021). This discrepancy suggests that the deep CMIP6 models are not saturated in land heat uptake and would take up even more energy if they were deeper. We employ the HFMod heat conduction forward model (see Section 2.3) that allows us to define the relationship between LHUe and land model depth in limited and infinitely deep land model configurations (Figure 1d). Since there is no official information on soil parameters for all CMIP6 models, we run the HFMod with parameter values for volumetric heat capacity and thermal diffusivity (Figure S1 in Supporting Information S1) over a range of common values of $0.5-1.5 \cdot 10^{-6}$ m⁻² s⁻¹ for the thermal diffusivity, and $2.5-3.5 \cdot 10^{-6}$ m⁻² s⁻¹ for the thermal diffusivity. 10⁶ Jm⁻³ K⁻¹ for the volumetric heat capacity (e.g., Cuesta-Valero et al., 2023). The results (Figure 1d; shading) illustrate that deviations in the effectiveness of heat conduction across models can govern a range of LHUe that captures the values of the deep CMIP6 model group. With a larger LSM depth and thus increased land heat uptake capacity, MPI-ESM SHALLOW and DEEP simulations have the same LHUe as the deep CMIP6 model average. MPI-ESM DEEP has slightly lower LHUe than the HFMod solution, consistent with lower global average values of $0.79 \cdot 10^{-6} \text{ m}^{-2} \text{ s}^{-1}$ for the thermal diffusivity and $2.0 \cdot 10^{6} \text{ Jm}^{-3} \text{ K}^{-1}$ for the volumetric heat capacity for MPI-ESM (Steinert, González-Rouco, Melo-Aguilar, et al., 2021). Any deviations of the CMIP6 model results from the solution of the heat conduction forward model might be additionally influenced by their consideration of soil moisture for the land heat uptake calculations, whereas the heat conduction forward model only considers thermal diffusion. The results confirm that LHUe saturates at 17.9 ZJ/K with a depth of about 200 m, where the diffusion of surface temperature anomalies with depth is attenuated completely.

4. Redistribution of Earth System Energy

Given the more realistic land heat uptake in simulations with a deeper land model, the question is how this affects the overall Earth energy budget and distribution among the climate system components. In the case where the energy is underestimated in the land system in the shallow LSM configurations, the assumption is that this energy is incorrectly distributed in other Earth system components, namely the ocean, atmosphere, and cryosphere.

Across all climate system components, the spread of the subsystem cumulative heat uptake between 1850 and 2100 in the CMIP6 models is relatively large (Figure 2). As expected, the ocean heat uptake is largest with 2828 ZJ (1609–3621 ZJ). As stated above, the land heat storage shows the largest uncertainty as a consequence of the depth limitations of some models and stores 56 ZJ (6–143 ZJ) across all models. Various ESMs match historical ocean heat uptake well, while cross-model uncertainty remains large (Table S3 in Supporting Information S1). The differences between the shallow and deep model configurations are most prominent in the land heat



Geophysical Research Letters



Figure 2. Cumulative heat uptake [ZJ] of 33 CMIP6 models (see Table S1 in Supporting Information S1) and the two MPI-ESM setups DEEP and SHALLOW for (a) ocean, (b) land, (c) atmosphere, (d) cryosphere (sea ice, land ice, snow), (e) the Earth Heat Content (EHC), which is the sum of all components in (a–d), and (f) integrated top of the atmosphere (TOA) energy imbalance in simulations between 1850 and 2100 following SSP5-85 forcing. For panel (e), EHC is only shown for a subset of models that allowed the calculation of all heat uptake components. For comparison, observational estimates (purple) from von Schuckmann et al. (2023) were added between 1960 and 2020. RMSE values are provided between the model-derived and observational estimates. Note the differences in *y*-axis between the panels.

uptake. MPI-ESM SHALLOW and DEEP align well with the CMIP6 averages, but the low ECS of MPI-ESM prevents more intense heat accumulation despite its LSM depth being sufficiently larger. Nonetheless, both the deep CMIP6 model group and MPI-ESM DEEP match observational estimates (added as relative changes to the CMIP6 ensemble mean in 1960) between 1960 and 2020 (Figure 2; von Schuckmann et al., 2023) with a difference of -6.6 and -2.9 ZJ at 2020, respectively, whereas the shallow models show considerably lower values with a difference of -17.4 and -16.9 ZJ at 2020, respectively. Here, the CMIP6 shallow model range of relative cumulative heat uptake by 2020 encompasses -0.2-20.4 ZJ, while the CMIP6 deep models range 13.9-23.2 ZJ--in comparison to the observed 23.9 ZJ. The atmosphere and cryosphere show the smallest heat uptake of 25 ZJ (15-37 ZJ) and 21 ZJ (9-28 ZJ), respectively, between 1850 and 2100. The spread of results for the cryosphere is partly caused by the lack of data availability (see Table S2 in Supporting Information S1), with some model output repositories missing variables to calculate the energy storage in some of the subcomponents of the cryosphere, likely causing the inconsistency with the observations. For example, the CMIP6 version of MPI-ESM does not include soil freezing and thus does not consider the contribution of energy from soil ice melt. Additionally, many models lack a dynamic representation of glaciers and ice caps, thus not including the heat exchange from their melting or growth. Although this can be considered a limitation of our analysis, the small contribution of the cryosphere heat uptake will likely not change the results considerably.

Solely considering absolute changes in the energy inventory can lead to inconsistencies in the effect of land model depth-related changes on the global heat distribution (Figures S3 and S4 in Supporting Information S1) due to model structural differences, internal variability, or model drift and the associated magnitude of energy conservation in many CMIP6 models (Irving et al., 2020). We therefore consider the heat distribution of climate subsystems as a fraction of the total Earth Heat Content (EHC) instead of the top of the atmosphere (TOA) energy imbalance (Cuesta-Valero et al., 2021). Due to the reported inconsistencies between the global energy budget and conservation (Irving et al., 2020; Wilde & Mulholland, 2020), we consider the EHC as the sum of heat contents in all climate system components.

For the shallow CMIP6 model group, our calculations find a fractional heat uptake distribution of 97.4% for the ocean, 1.1% for the land, and 0.8% and 0.6% for the atmosphere and the cryosphere, respectively (Figures 3a–3d). Note that only a subset of model experiments from the original CMIP6 ensemble is used here (17/20 shallow, and 9/13 deep models) based on data availability and excluding outliers that exceeded two standard deviations of heat



Geophysical Research Letters



Figure 3. Cumulative heat uptake for (a–d) global heat uptake in reference to EHC [% of EHC] of land, ocean, atmosphere, cryosphere (sea ice, land ice, snow), respectively, separated into shallow and deep model groups according to their land model depth (Table S1 in Supporting Information S1) for the period 1850–2100, and (e) the temporal evolution of the land and ocean heat uptake fractions [% of EHC] over the 21st century.

uptake percentages from the mean. The heat uptake partitioning highlights the limitations of a compromised land heat sink deviating substantially from the expected 5%-6% (von Schuckmann et al., 2020, 2023). The comparison between the deep and shallow CMIP6 model groups shows a considerable increase in percentages of the land heat uptake from 1.1% to 3.6% for the period 1850-2100. While the energy budgets of the atmosphere and the cryosphere stay relatively the same, the ocean loses heat fraction in the deep CMIP6 model group by about 1.5%. Fractional ocean (land) heat uptake ranges 95.5%–98.5% (0.3%–2.7%) and 93.5%–98.9% (2.6%–5.4%) for the 5th–95th percentiles in the shallow and deep CMIP sub-ensemble, respectively. The mean differences between the sub-ensembles of shallow and deep ESMs are statistically significant for the land and ocean heat uptake (t-test with p-value <0.05). These results are consistent with the heat content changes obtained by comparison of the MPI-ESM SHALLOW and DEEP simulations. Land and ocean heat uptake changes show similar magnitude in the DEEP model by about 2.0% (changing from 1.0% to 3.0%) and 2.0% (changing from 98.0% to 96.0%), respectively. In contrast to the CMIP6 ensemble, the DEEP MPI-ESM shows no decrease of heat in the cryosphere, likely due to the limitation in the parameterization of cryospheric elements in MPI-ESM. Land heat uptake remains slightly lower than the observational estimates, likely due to the dominating ocean heat uptake in both the CMIP6 and MPI-ESM ensembles, but consistent with the uncertainty in the observational estimate (Cuesta-Valero et al., 2021; von Schuckmann et al., 2023). Compared to the EHC-based estimates, taking the TOA imbalance as a reference for the global heat uptake between the shallow and deep CMIP6 model groups provides very similar results for the direction and magnitude of change of the land, atmosphere and cryosphere heat uptake fractions, while the changes of the ocean heat uptake are smaller (Figure S2 in Supporting Information S1). For TOA-based estimates, only land heat uptake change is statistically significant. We note that these results are consistent with the analysis of CMIP5 models over the historical period (Cuesta-Valero et al., 2021), indicating that, with respect to the main driver of land heat uptake - the land model depth, there has been no apparent improvement between the last and the current iteration of CMIP models (Cuesta-Valero et al., 2016).

Considering the global heat distribution over time also allows the assessment of temporal changes of the relative Earth system heat distribution (Figure 3e). The results show heat content percentages that are relatively constant over time. Larger variations at the beginning of the period are expected as the absolute amount of heat exchange in the Earth system is relatively small and can generate larger fractional changes. As the absolute Earth system heat content increases in the 21st century under the SSP5-85 scenario, the relative heat uptake of the ocean and land stabilizes. Given this stabilization of the global heat fractions, it is expected that the present-day ratios of global

heat uptake will not change substantially by the end of the 21st century. As most of the cryosphere is frozen water, its heat is expected to be transferred to the ocean and the atmosphere as part of a changing hydrological cycle under warming.

5. Summary and Conclusions

This study evaluates estimates of the land heat uptake from climate models with deep and shallow land model components under transient climate conditions and assesses their resulting influence on the global heat distribution. Since the average of the current generation of CMIP6 models does not realistically account for the land heat content in the Earth system (Cuesta-Valero et al., 2021; von Schuckmann et al., 2020), we separate them into two groups of shallow and deep models according to their land model depth and compare them to a set of simulations with MPI-ESM that uses two corresponding setups of a shallow and a deep land model component.

A clear influence of land model depth on the Earth heat inventory in Earth system models has been established. Deeper climate models generate Earth system heat distribution more consistent with observational and numerical estimates than the group of shallower models. The deeper group of CMIP6 models with depths between 42 and 90 m has a more realistic land heat uptake with improvements over the shallow model group. Furthermore, comparing shallow and deep CMIP6 land models with the standard shallow and a sufficiently deep MPI-ESM setups facilitates direct insight into the changes in the Earth system heat distribution. The results illustrate that the ocean can lose a fraction of the global heat budget to the land due to the increased land heat sink capacity. In both the CMIP6 models and MPI-ESM, the impacts of changes in the land heat uptake on the global heat inventory are relatively small because of the large natural variations of ocean heat content given its massive mass, thermal capacity and heat distribution capability through water advection (Levitus et al., 2000). The results for the ocean and land from climate models with deeper land model components are in closer agreement to the most recent observational estimates (von Schuckmann et al., 2023) than those with shallow components. Changes in the heat uptake fraction of the atmosphere and cryosphere are negligible and are partially sensitive to the representation of dynamic cryospheric elements in current-generation ESMs. Throughout the 21st century, the ratios of heat uptake between the climate subsystems stay relatively constant, even for a strong climate change scenario. Because energetic model biases in absolute terms tend to be larger between models than between intra-model climate subsystems (Cuesta-Valero et al., 2016, 2021; Irving et al., 2020; Wilde & Mulholland, 2020), individual impacts of a misrepresented energy distribution in the climate system may vary between models. Nevertheless, Earth energy distribution differences of a few percent of energy uptake may influence the terrestrial carbon cycle, hydro-thermodynamic coupling and biogeophysical properties of the land surface (Cuesta-Valero et al., 2023; Gulev et al., 2021).

Our results confirm that deeper models across CMIP6 are necessary to have a realistic representation of the land heat content and the corresponding subsurface thermodynamic state. We conclude that around two-thirds of CMIP6 models are compromised by an insufficiently deep land model depth. Further, we argue that even the deeper CMIP6 models are still not deep enough to capture the full potential for land heat storage since analytical solutions suggest deeper models of at least 170 m to be necessary (Steinert, González-Rouco, Melo-Aguilar, et al., 2021). Their land heat uptake efficacy comparable to the heat conduction model solutions likely stems from uncertainties in the consideration of soil thermal properties. Additional computational requirements from deeper LSMs may add up with the use of high-density layer discretization and increasing process fidelity, but generally remains small compared to the computational costs of other Earth system components (González-Rouco et al., 2021; Steinert, González-Rouco, Melo-Aguilar, et al., 2021). Overall, these results highlight the relevance of the land heat sink in simulating the global energy inventory under global warming.

Data Availability Statement

The calculation of the Earth heat inventories is based on CMIP6 data that is publicly available on servers provided by ESGF (https://esgf-node.llnl.gov/projects/cmip6/) and following the methods described in Cuesta-Valero et al. (2021). Observational estimates for the heat uptake of all Earth system component were taken from von Schuckmann et al. (2023) accessible under https://www.wdc-climate.de/ui/entry?acronym=GCOS_EHI_1960-2020. Data of the modified MPI-ESM experiments used herein are available from Zenodo (Steinert et al., 2024).



Acknowledgments

This publication is supported by the SMILEME project (PID2021-126696OB-C21), funded by MCIN/AE/10.13039/ 501100011033 and iLESM (LINKB20082) funded by the iLINK program in CSIC. NJS acknowledges funding from the IlModels (CGL2014-59644-R) and GreatModelS (RTI2018-102305-B-C21) projects funded by the Spanish MINECO, as well as funding from the Norwegian Research Centre AS (NORCE). FJCV was partially funded as research fellow of the Alexander von Humboldt Foundation and as a fellow of the Climate Change Initiative (CCI) of the European Space Agency (ESA) at the Helmholtz Centre for Environmental Research (UFZ). FGP was funded by contract PRE2019-090694 of the Spanish Ministry of Science and Innovation. The MPI-ESM simulations, and model data storage were performed at the Deutsches Klima Rechenzentrum (DKRZ) using resources granted by its Scientific Steering Committee (WLA) under project ID bm1026.

References

- Alexeev, V. A., Nicolsky, D. J., Romanovsky, V. E., & Lawrence, D. M. (2007). An evaluation of deep soil configurations in the CLM3 for improved representation of permafrost. *Geophysical Research Letters*, 34(9), L09502. https://doi.org/10.1029/2007GL029536
- Burke, E. J., Zhang, Y., & Krinner, G. (2020). Evaluating permafrost physics in the Coupled Model Intercomparison Project 6 (CMIP6) models and their sensitivity to climate change. *The Cryosphere*, *14*(9), 3155–3174. https://doi.org/10.5194/tc-14-3155-2020
- Carslaw, H., & Jaeger, J. (1959). Conduction of heat in solids (2nd ed.). Clarendon Press.
- Champeaux, J. L., Masson, V., & Chauvin, F. (2005). ECOCLIMAP: A global database of land surface parameters at 1 km resolution. *Meteorological Applications*, 12(1), 29–32. https://doi.org/10.1017/s1350482705001519
- Cheruy, F., Ducharne, A., Hourdin, F., Musat, I., Vignon, É., Gastineau, G., et al. (2020). Improved near-surface continental climate in IPSL-CM6A-LR by combined evolutions of atmospheric and land surface physics. *Journal of Advances in Modeling Earth Systems*, 12(10), e2019MS002005. https://doi.org/10.1029/2019MS002005
- Courant, R., Friedrichs, K., & Lewy, H. (1928). Über die partiellen Differenzengleichungen der mathematischen Physik. *Mathematische Annalen*, 100(1), 32–74. https://doi.org/10.1007/bf01448839
- Cuesta-Valero, F. J., Beltrami, H., García-García, A., Krinner, G., Langer, M., MacDougall, A. H., et al. (2023). Continental heat storage: Contributions from the ground, inland waters, and permafrost thawing. *Earth System Dynamics*, 14(3), 609–627. https://doi.org/10.5194/esd-14-609-2023
- Cuesta-Valero, F. J., García-García, A., Beltrami, H., González-Rouco, J. F., & García-Bustamante, E. (2021). Long-term global ground heat flux and continental heat storage from geothermal data. *Climate of the Past*, 17(1), 451–468. https://doi.org/10.5194/cp-17-451-2021
- Cuesta-Valero, F. J., García-García, A., Beltrami, H., & Smerdon, J. E. (2016). First assessment of continental energy storage in CMIP5 simulations. *Geophysical Research Letters*, 43(10), 5326–5335. https://doi.org/10.1002/2016GL068496
- Dutton, J. A. (2002). The ceaseless wind: An introduction to the theory of atmospheric motion. Dover Publications.
- Fisher, R. A., & Koven, C. D. (2020). Perspectives on the future of land surface models and the challenges of representing complex terrestrial systems. Journal of Advances in Modeling Earth Systems, 12(4), e2018MS001453. https://doi.org/10.1029/2018MS001453
- Flato, G., Marotzke, J., Abiodun, B., Braconnot, P., Chou, S., Collins, W., et al. (2013). Evaluation of climate models. In T. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), *Climate change 2013: The physical science basis. contribution of working group i to the fifth assessment report of the intergovernmental panel on climate change* (pp. 741–866). Cambridge University Press. https://doi.org/10. 1017/CB09781107415324.020
- García-Pereira, F., González-Rouco, J. F., Melo-Aguilar, C., Steinert, N. J., García-Bustamante, E., de Vrese, P., et al. (2024). First comprehensive assessment of industrial era land heat uptake from multiple sources. *Earth System Dynamics Discussions*. https://doi.org/10.5194/esd-2023-44
- Gidden, M. J., Riahi, K., Smith, S. J., Fujimori, S., Luderer, G., Kriegler, E., et al. (2019). Global emissions pathways under different socioeconomic scenarios for use in CMIP6: A dataset of harmonized emissions trajectories through the end of the century. *Geoscientific Model* Development, 12(4), 1443–1475. https://doi.org/10.5194/gmd-12-1443-2019
- González-Rouco, J. F., Beltrami, H., Zorita, E., & Stevens, M. B. (2009). Borehole climatology: A discussion based on contributions from climate modeling. *Climate of the Past*, 5(1), 97–127. https://doi.org/10.5194/cp-5-97-2009
- González-Rouco, J. F., Steinert, N. J., García-Bustamante, E., Hagemann, S., de Vrese, P., Jungclaus, J. H., et al. (2021). Increasing the depth of a Land Surface Model. Part I: Impacts on the soil thermal regime and energy storage. *Journal of Hydrometeorology*. https://doi.org/10.1175/ JHM-D-21-0024.1
- Gornall, J., Betts, R., Burke, E., Clark, R., Camp, J., Willett, K., & Wiltshire, A. (2010). Implications of climate change for agricultural productivity in the early twenty-first century. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 365(1554), 2973–2989. https://doi.org/10.1098/rstb.2010.0158
- Griffies, S. (2004). Fundamentals of ocean climate models. Princeton University Press.
- Gulev, S., Thorne, P., Ahn, J., Dentener, F., Domingues, C., Gerland, S., et al. (2021). Changing state of the climate system. In Climate change 2021 – the physical science basis: Working group I contribution to the sixth assessment report of the intergovernmental panel on climate change (pp. 287–422). Cambridge University Press.
- Gupta, A. S., Jourdain, N. C., Brown, J. N., & Monselesan, D. (2013). Climate drift in the CMIP5 models. Journal of Climate, 26(21), 8597–8615. https://doi.org/10.1175/JCLI-D-12-00521.1
- Hansen, J., Nazarenko, L., Ruedy, R., Sato, M., Willis, J., Genio, A. D., et al. (2005). Earth's energy imbalance: Confirmation and implications. *Science*, 308(5727), 1431–1435. https://doi.org/10.1126/science.1110252
- Hansen, J., Sato, M., Kharecha, P., & von Schuckmann, K. (2011). Earth's energy imbalance and implications. Atmospheric Chemistry and Physics, 11(24), 13421–13449. https://doi.org/10.5194/acp-11-13421-2011
- Hobbs, W., Palmer, M. D., & Monselesan, D. (2016). An energy conservation analysis of ocean drift in the CMIP5 global coupled models. Journal of Climate, 29(5), 1639–1653. https://doi.org/10.1175/JCLI-D-15-0477.1
- Irving, D., Hobbs, W., Church, J., & Zika, J. (2020). A mass and energy conservation analysis of drift in the CMIP6 ensemble. *Journal of Climate*, 34(8), 1–43. https://doi.org/10.1175/jcli-d-20-0281.1
- Lawrence, D. M., Fisher, R. A., Koven, C. D., Oleson, K. W., Swenson, S. C., Bonan, G., et al. (2019). The community land model version 5: Description of new features, benchmarking, and impact of forcing uncertainty. *Journal of Advances in Modeling Earth Systems*, 11(12), 4245–4287. https://doi.org/10.1029/2018MS001583
- Levitus, S., Antonov, J. I., Boyer, T. P., & Stephens, C. (2000). Warming of the World Ocean. Science, 287(5461), 2225–2229. https://doi.org/10. 1126/science.287.5461.2225
- MacDougall, A. H., Beltrami, H., González-Rouco, J. F., Stevens, M. B., & Bourlon, E. (2010). Comparison of observed and general circulation model derived continental subsurface heat flux in the Northern Hemisphere. *Journal of Geophysical Research*, 115(D12), D12109. https://doi. org/10.1029/2009JD013170
- MacDougall, A. H., González-Rouco, J. F., Stevens, M. B., & Beltrami, H. (2008). Quantification of subsurface heat storage in a GCM simulation. Geophysical Research Letters, 35(13), L13702. https://doi.org/10.1029/2008GL034639
- Mauritsen, T., Bader, J., Becker, T., Behrens, J., Bittner, M., Brokopf, R., et al. (2019). Developments in the MPI-M Earth System Model version 1.2 (MPI-ESM1.2) and its response to increasing CO₂. Journal of Advances in Modeling Earth Systems, 11(4), 998–1038. https://doi.org/10. 1029/2018MS001400
- McDougall, T. J. (2003). Potential enthalpy: A conservative oceanic variable for evaluating heat content and heat fluxes. *Journal of Physical Oceanography*, 33(5), 945–963. https://doi.org/10.1175/1520-0485(2003)033(0945:peacov)2.0.co;2

- McDougall, T. J., Jackett, D. R., Wright, D. G., & Feistel, R. (2003). Accurate and computationally efficient algorithms for potential temperature and density of seawater. *Journal of Atmospheric and Oceanic Technology*, 20(5), 730–741. https://doi.org/10.1175/1520-0426(2003)20(730: aaceaf)2.0.co;2
- Meehl, G. A., Senior, C. A., Eyring, V., Flato, G., Lamarque, J. F., Stouffer, R. J., et al. (2020). Context for interpreting equilibrium climate sensitivity and transient climate response from the CMIP6 Earth system models. *Science Advances*, 6(26), eaba1981. https://doi.org/10.1126/ sciadv.aba1981
- Meinshausen, M., Vogel, E., Nauels, A., Lorbacher, K., Meinshausen, N., Etheridge, D. M., et al. (2017). Historical greenhouse gas concentrations for climate modelling (CMIP6). *Geoscientific Model Development*, 10(5), 2057–2116. https://doi.org/10.5194/gmd-10-2057-2017
- Oleson, K. W., Lawrence, D. M., Bonan, B. G., Drewniak, B., Huang, M., Koven, C. D., et al. (2013). Technical description of version 4.5 of the community land model (CLM). https://doi.org/10.5065/D6RR1W7M
- Oleson, K. W., Lawrence, D. M., Bonan, G. B., Flanner, M. G., Kluzek, E., Lawrence, P. J., et al. (2010). Technical description of version 4.0 of the community land model (CLM) (Tech. Rep. Nos. NCAR/TN-478+STR). University Corporation for Atmospheric Research. https://doi.org/ 10.5065/D6FB50WZ
- Peacock, S., & Maltrud, M. (2006). Transit-Time distributions in a global ocean model. Journal of Physical Oceanography, 36(3), 474–495. https://doi.org/10.1175/JPO2860.1
- Phillips, C. L. (2020). How much will soil warm? Journal of Geophysical Research: Biogeosciences, 125(7), e2020JG005668. https://doi.org/10. 1029/2020JG005668
- Previdi, M., Smith, K. L., & Polvani, L. M. (2015). How well do the CMIP5 models simulate the antarctic atmospheric energy budget? *Journal of Climate*, 28(20), 7933–7942. https://doi.org/10.1175/jcli-d-15-0027.1
- Reick, C. H., Gayler, V., Goll, D., Hagemann, S., Heidkamp, M., Nabel, J. E. M. S., et al. (2021). JSBACH 3 The land component of the MPI Earth System Model: documentation of version 3.2. Berichte zur Erdsystemforschung (Technical Report No. 240). MPI f
 ür Meteorologie. https://doi.org/10.17617/2.3279802
- Rhein, M., Rintoul, S., Aoki, S., Campos, E., Chambers, D., Feely, R., et al. (2013). Observations: Ocean [Book Section]. In T. Stocker, G.-K. Qin, M. Plattner, S. K. Tignor, J. Allen, J. Boschung, et al. (Eds.), *Climate change 2013: The physical science basis. Contribution of working group I* to the fifth assessment report of the intergovernmental panel on climate change (pp. 255–316). Cambridge University Press. https://doi.org/10. 1017/CBO9781107415324.010
- Roache, P. J. (1998). Computational fluid dynamics (2nd ed.). Hermosa.
- Schlund, M., Lauer, A., Gentine, P., Sherwood, S. C., & Eyring, V. (2020). Emergent constraints on equilibrium climate sensitivity in CMIP5: Do they hold for CMIP6? *Earth System Dynamics*, 11(4), 1233–1258. https://doi.org/10.5194/esd-11-1233-2020
- Séférian, R., Gehlen, M., Bopp, L., Resplandy, L., Orr, J. C., Marti, O., et al. (2016). Inconsistent strategies to spin up models in CMIP5: Implications for ocean biogeochemical model performance assessment. *Geoscientific Model Development*, 9(5), 1827–1851. https://doi.org/10. 5194/gmd-9-1827-2016
- Smerdon, J. E., & Stieglitz, M. (2006). Simulating heat transport of harmonic temperature signals in the Earth's shallow subsurface: Lowerboundary sensitivities. *Geophysical Research Letters*, 33(14), L14402. https://doi.org/10.1029/2006GL026816
- Smith, R., Jones, P., Briegleb, B., Bryan, F., Danabasoglu, G., Dennis, J., et al. (2010). The Parallel Ocean Program (POP) reference manual. Ocean Component of the Community Climate System Model (CCSM) and Community Earth System Model (CESM) (Technical report no. LAUR-01853). National Center for Atmospheric Research. https://doi.org/10.5065/D6FB50WZ
- Steele, M., Morley, R., & Ermold, W. (2001). PHC: A global ocean hydrography with a high-quality Arctic Ocean. Journal of Climate, 14(9), 2079–2087. https://doi.org/10.1175/1520-0442(2001)014(2079:pagohw)2.0.co;2
- Steinert, N. J., Cuesta-Valero, F. J., García-Pereira, F., de Vrese, P., Melo Aguilar, C. A., García-Bustamante, E., et al. (2024). Dataset: Underestimated land heat uptake alters the global energy distribution in CMIP6 climate models [Dataset]. Zenodo. https://doi.org/10.5281/ zenodo.10669977
- Steinert, N. J., González-Rouco, J. F., de Vrese, P., García-Bustamante, E., Hagemann, S., Melo-Aguilar, C., et al. (2021). Increasing the depth of a Land Surface Model. Part II: Sensitivity to improved coupling between soil hydrology and thermodynamics and associated permafrost response. Journal of Hydrometeorology. https://doi.org/10.1175/JHM-D-21-0023.1
- Steinert, N. J., González-Rouco, J. F., Melo-Aguilar, C., García-Pereira, F., García-Bustamante, E., de Vrese, P., et al. (2021). Agreement of analytical and simulation-based estimates of the required land depth in climate models. *Geophysical Research Letters*, 48(20), e2021GL094273. https://doi.org/10.1029/2021GL094273
- Stevens, M. B., Smerdon, J. E., González-Rouco, J. F., Stieglitz, M., & Beltrami, H. (2007). Effects of bottom boundary placement on subsurface heat storage: Implications for climate model simulations. *Geophysical Research Letters*, 34(2), L02702. https://doi.org/10.1029/ 2006GL028546
- Trenberth, K. E. (1997). Using atmospheric budgets as a constraint on surface fluxes. Journal of Climate, 10(11), 2796–2809. https://doi.org/10. 1175/1520-0442(1997)010(2796:uabaac)2.0.co;2
- Van Wijk, W. R., Borghrst, A. J. W., Businger, J. A., Derksen, W. J., Schmidt, F. H., Scholte Ubing, D. W., & De Vries, D. A. (1963). *Physics of plant environment*. North-Holland Publishing Company.
- von Schuckmann, K., Cheng, L., Palmer, M. D., Hansen, J., Tassone, C., Aich, V., et al. (2020). Heat stored in the Earth system: Where does the energy go? *Earth System Science Data*, *12*(3), 2013–2041. https://doi.org/10.5194/ESSD-12-2013-2020
- von Schuckmann, K., Minière, A., Gues, F., Cuesta-Valero, F. J., Kirchengast, G., Adusumilli, S., et al. (2023). Heat stored in the Earth system 1960–2020: Where does the energy go? *Earth System Science Data*, *15*(4), 1675–1709. https://doi.org/10.5194/essd-15-1675-2023
- Wilde, S. P. R., & Mulholland, P. (2020). An analysis of the Earth's energy budget. International Journal of Atmospheric and Oceanic Sciences, 4(2), 54. https://doi.org/10.11648/j.ijaos.20200402.12
- Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi, P., et al. (2020). Causes of higher climate sensitivity in CMIP6 models. *Geophysical Research Letters*, 47(1), e2019GL085782. https://doi.org/10.1029/2019gl085782

References From the Supporting Information

Andrews, M. B., Ridley, J. K., Wood, R. A., Andrews, T., Blockley, E. W., Booth, B., et al. (2020). Historical simulations with HadGEM3-GC3.1 for CMIP6. *Journal of Advances in Modeling Earth Systems*, 12(6), e2019MS001995. https://doi.org/10.1029/2019ms001995

Bi, D., Dix, M., Marsland, S., O'Farrell, S., Sullivan, A., Bodman, R., et al. (2020). Configuration and spin-up of ACCESS-CM2, the new generation Australian community climate and Earth system simulator coupled model. *Journal of Southern Hemisphere Earth Systems Science*, 70(1), 225–251. https://doi.org/10.1071/es19040

19448007, 2024, 10, Downlo

from https:

elibrary.wiley

- Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov, V., et al. (2020). Presentation and evaluation of the IPSL-CM6A-LR climate model. *Journal of Advances in Modeling Earth Systems*, 12(7), e2019MS002010. https://doi.org/10.1029/2019ms002010 Cherchi, A., Fogli, P. G., Lovato, T., Peano, D., Iovino, D., Gualdi, S., et al. (2019). Global mean climate and main patterns of variability in the
- CMCC-CM2 coupled model. Journal of Advances in Modeling Earth Systems, 11(1), 185–209. https://doi.org/10.1029/2018ms001369
 Danabasoglu, G., Lamarque, J., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J., et al. (2020). The Community Earth System Model
- Version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12(2), e2019MS001916. https://doi.org/10.1029/2019ms001916
 Döscher, R., Acosta, M., Alessandri, A., Anthoni, P., Arsouze, T., Bergman, T., et al. (2021). The EC-Earth3 Earth system model for the Coupled Model Intercomparison Project 6. Geoscientific Model Development, 15(7), 2973–3020. https://doi.org/10.5194/gmd-15-2973-2022
- Dunne, J. P., Horowitz, L. W., Adcroft, A. J., Ginoux, P., Held, I. M., John, J. G., et al. (2020). The GFDL Earth System Model Version 4.1 (GFDL-ESM 4.1): Overall Coupled Model description and simulation characteristics. *Journal of Advances in Modeling Earth Systems*, 12(11), e2019MS002015. https://doi.org/10.1029/2019ms002015
- Golaz, J., Caldwell, P. M., Roekel, L. P. V., Petersen, M. R., Tang, Q., Wolfe, J. D., et al. (2019). The DOE E3SM Coupled Model version 1: Overview and evaluation at standard resolution. *Journal of Advances in Modeling Earth Systems*, 11(7), 2089–2129. https://doi.org/10.1029/ 2018ms001603
- Hajima, T., Watanabe, M., Yamamoto, A., Tatebe, H., Noguchi, M. A., Abe, M., et al. (2020). Development of the MIROC-ES2L Earth system model and the evaluation of biogeochemical processes and feedbacks. *Geoscientific Model Development*, 13(5), 2197–2244. https://doi.org/10. 5194/gmd-13-2197-2020
- He, B., Bao, Q., Wang, X., Zhou, L., Wu, X., Liu, Y., et al. (2019). CAS FGOALS-f3-L model datasets for CMIP6 historical atmospheric model Intercomparison project simulation. Advances in Atmospheric Sciences, 36(8), 771–778. https://doi.org/10.1007/s00376-019-9027-8
- Held, I. M., Guo, H., Adcroft, A., Dunne, J. P., Horowitz, L. W., Krasting, J., et al. (2019). Structure and performance of GFDL's CM4.0 climate model. *Journal of Advances in Modeling Earth Systems*, 11(11), 3691–3727. https://doi.org/10.1029/2019ms001829
- Kelley, M., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Ruedy, R., Russell, G. L., et al. (2020). GISS-E2.1: Configurations and climatology. Journal of Advances in Modeling Earth Systems, 12(8), e2019MS002025. https://doi.org/10.1029/2019ms002025
- Li, L., Yu, Y., Tang, Y., Lin, P., Xie, J., Song, M., et al. (2020). The Flexible Global Ocean-Atmosphere-Land System Model Grid-Point version 3 (FGOALS-g3): Description and evaluation. *Journal of Advances in Modeling Earth Systems*, 12(9), e2019MS002012. https://doi.org/10.1029/ 2019ms002012
- Lovato, T., Peano, D., Butenschön, M., Materia, S., Iovino, D., Scoccimarro, E., et al. (2022). CMIP6 simulations with the CMCC Earth System Model (CMCC-ESM2). Journal of Advances in Modeling Earth Systems, 14(3). https://doi.org/10.1029/2021ms002814
- Séférian, R., Nabat, P., Michou, M., Saint-Martin, D., Voldoire, A., Colin, J., et al. (2019). Evaluation of CNRM Earth System Model, CNRM-ESM2-1: Role of Earth system processes in present-day and future climate. *Journal of Advances in Modeling Earth Systems*, 11(12), 4182–4227. https://doi.org/10.1029/2019ms001791
- Seland, Ø., Bentsen, M., Olivié, D., Toniazzo, T., Gjermundsen, A., Graff, L. S., et al. (2020). Overview of the Norwegian Earth System Model (NorESM2) and key climate response of CMIP6 DECK, historical, and scenario simulations. *Geoscientific Model Development*, 13(12), 6165–6200. https://doi.org/10.5194/gmd-13-6165-2020
- Sellar, A. A., Walton, J., Jones, C. G., Wood, R., Abraham, N. L., Andrejczuk, M., et al. (2020). Implementation of U.K. Earth System Models for CMIP6. Journal of Advances in Modeling Earth Systems, 12(4), e2019MS001946. https://doi.org/10.1029/2019ms001946
- Swart, N. C., Cole, J. N. S., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., et al. (2019). The Canadian Earth System Model version 5 (CanESM5.0.3). Geoscientific Model Development, 12(11), 4823–4873. https://doi.org/10.5194/gmd-12-4823-2019
- Tatebe, H., Ogura, T., Nitta, T., Komuro, Y., Ogochi, K., Takemura, T., et al. (2019). Description and basic evaluation of simulated mean state, internal variability, and climate sensitivity in MIROC6. Geoscientific Model Development, 12(7), 2727–2765. https://doi.org/10.5194/gmd-12-2727-2019
- Voldoire, A., Saint-Martin, D., Sénési, S., Decharme, B., Alias, A., Chevallier, M., et al. (2019). Evaluation of CMIP6 DECK experiments with CNRM-CM6-1. Journal of Advances in Modeling Earth Systems, 11(7), 2177–2213. https://doi.org/10.1029/2019ms001683
- Wang, Y., Hsu, H., Chen, C., Tseng, W., Hsu, P., Lin, C., et al. (2021). Performance of the Taiwan Earth System Model in simulating climate variability compared with observations and CMIP6 model simulations. *Journal of Advances in Modeling Earth Systems*, 13(7), e2020MS002353. https://doi.org/10.1029/2020ms002353
- Wu, T., Lu, Y., Fang, Y., Xin, X., Li, L., Li, W., et al. (2019). The Beijing Climate Center Climate System Model (BCC-CSM): The main progress from CMIP5 to CMIP6. Geoscientific Model Development, 12(4), 1573–1600. https://doi.org/10.5194/gmd-12-1573-2019
- Xin-Yao, R., JianCHEHao-Ming, L., Yu-Fei, X., Jing-Zhi, S., Li-Juan, H., & Zheng-Qiu, Z. (2019). Introduction of CAMS-CSM model and its participation in CMIP6. *Climate Change Research*, 15(5), 540–544. https://doi.org/10.12006/j.issn.1673-1719.2019.186
- Yukimoto, S., Kawai, H., Koshiro, T., Oshima, N., Yoshida, K., Urakawa, S., et al. (2019). The Meteorological Research Institute Earth System Model version 2.0, MRI-ESM2.0: Description and basic evaluation of the physical component. *Journal of the Meteorological Society of Japan.* Ser. 11, 97(5), 931–965. https://doi.org/10.2151/jmsj.2019-051
- Zhang, H., Zhang, M., Jin, J., Fei, K., Ji, D., Wu, C., et al. (2020). Description and climate simulation performance of CAS-ESM version 2. *Journal of Advances in Modeling Earth Systems*, *12*(12), e2020MS002210. https://doi.org/10.1029/2020ms002210